

International Journal of Pharmacology

ISSN 1811-7775





∂ OPEN ACCESS

International Journal of Pharmacology

ISSN 1811-7775 DOI: 10.3923/ijp.2024.630.635



Research Article Impact of Metformin on Hypothyroidism Induced by Cyclophosphamide in Rats

Ahmad Hamad Alhowail and Maha Abdulrahman Aldubayan

Department of Pharmacology and Toxicology, College of Pharmacy, Qassim University, Kingdom of Saudi Arabia

Abstract

Background and Objective: Cyclophosphamide (CYP), a commonly used chemotherapeutic agent, is administered to patients to treat various cancer types. The toxic effects of this substance are widely known and can cause damage to the liver, kidneys, nervous system and bone marrow. Metformin (MET) is frequently prescribed medication for diabetes mellitus type-2, known for its effectiveness in managing blood sugar levels. Present research aims to study the effect of co-administering MET on CYP-induced hypothyroidism by determining the level of thyroid hormones. Materials and Methods: Forty rats, with a body weight ranging from 220 to 250 g, were divided into four groups of ten animals, each for control and treatment purposes. Saline was given to animals in control group. The CYPtreated group was administered four doses of CYP (100 mg/kg each) by intraperitoneal route. The MET administered animals received 3 mg/mL dose mixed in their drinking water, starting one day preceding to the CYP jab and continued until the last dose. Further, the combination group of animals was administered four doses of a CYP and a daily MET through their drinking water. Daily observations were made on the animals for any signs of mortality, while blood samples were collected to measure TSH as well as thyroid hormones (free and total T3 and T4 concentrations). **Results:** The study's data suggested that the administration of CYP resulted in a higher mortality rate than the control animals. Additionally, when combined with MET, CYP further decreased the survival rate. Furthermore, a significant reduction in the concentrations of free and total T3 and T4 were observed in animals treated with CYP and CYP+MET upon comparison with control group. On the other hand, no significant changes in TSH level were recorded across all groups. Conclusion: The findings of the study revealed that the induction of hypothyroidism by CYP was not reduced by co-treatment with MET. This suggests that while MET does have anti-proliferative properties but the toxic effects of CYP was augmented, when used together.

Key words: Cyclophosphamide, hypothyroidism, metformin, thyroid, rats

Citation: Alhowail, A.H. and M.A. Aldubayan, 2024. Impact of metformin on hypothyroidism induced by cyclophosphamide in rats. Int. J. Pharmacol., 20: 630-635.

Corresponding Author: Ahmad Hamad Alhowail, Department of Pharmacology and Toxicology, College of Pharmacy, Qassim University, Kingdom of Saudi Arabia

Copyright: © 2024 Ahmad Hamad Alhowail and Maha Abdulrahman Aldubayan. This is an open access article distributed under the terms of the creative commons attribution License, which permits unrestricted use, distribution and reproduction in any medium, provided the original author and source are credited.

Competing Interest: The authors have declared that no competing interest exists.

Data Availability: All relevant data are within the paper and its supporting information files.

INTRODUCTION

Thyroid hormones such as T3 (Triiodothyronine) and T4 (Thyroxine) is known to control various physiological functions¹. Their levels are crucial for the development and functioning of the body¹. The thyroid gland produces T3 and T4, which are then released into the bloodstream². These hormones can travel to all organs, including the brain and binds to thyroid hormone receptors (THRs)³. The THRs are classified as nuclear receptors and exists in two isoforms (THR α and THR β) with varying distributions in different tissues⁴. According to previous studies, THRa1 is mostly located in heart and skeletal muscle, while liver, kidneys and brain predominantly contains THR⁵. Imbalances in the thyroid hormones and THRs expressions can disrupt normal bodily functions⁵. It is clear that mice with THRα-knockout in the hippocampus show a cognitive deficiency, providing further evidence of the importance of $THR\alpha$ in brain and cognitive function⁶. Thus, hypothyroidism is marked by irregular concentrations of thyroid hormones, which in turn affect various organs, including the receptors for THRs, leading to changes in their functioning⁷. Neuroimaging studies have further identified variations in brain structure and function in hypothyroidism diagnosed patients⁸.

Cyclophosphamide commonly abbreviated as CYP is an anticancer medication obtained by synthetic reaction. This drug is an alkylating agent belonging to the nitrogen mustard class⁹. It is the liver that is responsible for the activation of CYP, which ultimately leads to the transformation of mustard into alkylating phosphoramide¹⁰. The drug accomplishes its action by alkyl group affixation to the DNA guanine base, which ultimately leads to the formation of cross-links that are irreversible in the DNA strands¹¹. In the end, this results in the death of cells that is mostly observed at S and G2 cell cycle phases¹². Because of its cytotoxic effects, it is widely utilized in the treatment of various types of carcinomas, either on its own or in conjunction with other medications¹². At lower doses, CYP is also prescribed for certain autoimmunity conditions (multiple sclerosis, rheumatoid arthritis, glomerulonephritis, systemic lupus erythematosus) and as a preventive measure for organ rejection after transplantation^{13,14}. The treatment with CYP can lead to various negative effects, including hair loss, low platelet count, ulcers in the mucous membranes, changes in skin color, scarring of the lungs, facial injuries, low white blood cell count, blood in the urine, bleeding in the lungs and small intestine, as well as problems with thinking and an underactive thyroid¹⁵⁻¹⁷. These complications have been found to significantly impact the treatment's prognosis, often resulting in unfortunate outcomes.

Metformin (MET) is classified as an insulin sensitizer and is commonly prescribed to manage non-insulin dependent diabetes mellitus¹⁸. The medication effectively lowers blood sugar levels by targeting various mechanisms, including reducing glucose production in hepatic and enhancing glucose uptake in organs such as liver, intestine as well as muscles¹⁹. Reported pharmacological properties of MET include reducing fat deposits and anti-tumor cell proliferation, even in non-diabetic individuals¹⁸. Previous research has shown that combining MET with a well-established anticancer drug can have positive effects²⁰. Nevertheless, the studies have not provided any information regarding the impact of using CYP and MET in combination on thyroid function. Hence, present study aims to assess the role of MET on thyroid function in rats undergoing chronic treatment with CYP. The concentrations of T3 and T4 (free and total) as well as TSH were measured using Electrochemiluminescence (ECL) method.

MATERIALS AND METHODS

Study area: The study was carried out at the Pharmacology Laboratory in the College of Pharmacy, Qassim University, Saudi Arabia, from September 2019 to October, 2019.

Chemicals and drugs: Drugs required for the research were purchased from market such that Endoxan[®] (Baxter Ltd., Germany) and Metfor[®] (Tabuk Pharmaceutical Company of Saudi Arabia) were selected for cyclophosphamide and metformin hydrochloride, respectively.

Animals: A group of forty rats, all 10-12 weeks old, obtained from animal house in the College of Pharmacy, Qassim University, were kept in separate housing units within an environment that followed a light/dark cycle of 12 hours, with the illumination switching on at 6:00 in the morning. The animals were provided with unrestricted admittance to pelleted food and water in the 24 hrs. The experimental rats were observed each day for the survival rate. The studies were conducted following the necessary ethical procedures for animal experimentation, as required by the institution.

Experimental design and treatment protocol: For the purpose of this investigation, the animals were separated into four groups, with ten animals belonging to each group. Intraperitoneal administration of normal saline serves as the control. Treatment 1: CYP is to be administered

intraperitoneally at a concentration of 100 mg/kg, 4 times/day, on alternate days. The second treatment consisted of MET (3 mg/mL) dissolved in drinking water and induced for a period of 10 days. The fourth was a combination treatment, wherein MET was given for a period of 10 days and on the starting day of MET induction, CYP was administered in four doses on alternate days. Following the conclusion of the treatments, the animals were put to sleep and blood samples were taken. The experimental rats were monitored every day to ensure that their survival/mortality rates were accurately recorded.

Electrochemiluminescence (ECL): During the tenth day, after the rats had been put under hypnosis with Carbon Dioxide (CO_2) , they were put to euthanized in a humane manner by having their heads severed from their necks. This occurred following the last administration of the drug. Shortly after, blood samples were gathered from the control/untreated group and the treatment groups such as those received MET, CYP and CYP+MET. Samples collected were subsequently positioned into tubes having Ethylenediaminetetraacetic Acid (EDTA). After the mentioned procedure, the vials with the blood samples underwent centrifugation (12,000 \times g) for ten mins. This process led to the isolation of the plasma component. The levels of biomarkers such as TSH, T4, FT3 and T3 were measured by means of an automatic analyzer that relies on electrochemiluminescence (ECL) technique based on immunoassay concept. The task was carried out in compliance with the procedures described by the maker, Roche Diagnostics, headquartered in Germany²¹.

Statistical analysis: The values collected from the *in vivo* study underwent ANOVA (One-way Analysis of Variance) test.

The outcomes of the research were reported as the Mean \pm SEM (Standard Error of the Mean). Subsequently, a Tukey analysis was employed to compare the data individually, with statistically significance values being determined by a p-value below 0.05.

RESULTS

CYP and MET impact on the survival rate in rats: A equivalent rat method was established in order to gain a improved understanding of whether or not MET induced an effect on reducing the survival/endurance rate of rats that were administered with CYP. This was accomplished through the utilization of a Kaplan-Meier test. According to the findings of the survival rate analysis that was carried out between rats that were induced with CYP as well as rats that were treated with CYP and MET. The results suggested that the treatment with CYP and MET did, in fact, considerably reduce the survival rate when related to the treatment with CYP alone. Therefore, the findings of the study demonstrated that the inclusion of MET to CYP has the possibility to bring about the lethal effect of CYP. Each of the saline as well as MET groups maintained a survival rate of one hundred percent (Fig. 1).

Impact of CYP and MET on TSH concentrations: The findings of the ECL test revealed no discernible variation in the concentrations of TSH amid the groups that were given either saline or those that were given treatments. As a result, both MET as well as CYP treatments appear to had no observable consequence on the concentrations of TSH (Fig. 2).

Impact of CYP and MET on free and total T3 and T4 concentrations: Based on the results of the ECL study, the



Fig. 1: CYP and MET impact on the survival/endurance rate

Survival rate of the rats treated with CYP is decreased when combined with MET. The impact of MET on survival was assessed in rat models. The treatment regimen consisted of four dosages of CYP (400 mg/kg), while MET was given by oral route after it was dissolved into the drinking water on a continuous basis

Int. J. Pharmacol., 20 (4): 630-635, 2024



Fig. 2: Exploring the effects of CYP, MET and their combined usage on TSH levels Graph indicates that no notable variations were recorded among the four rats' groups in the study. The value is presented as the standard error with mean for six rats in each group and it was analyzed using a statistical analysis method



Fig. 3(a-d): Effects of CYP, MET and their combination on total (a) T4, (b) Free T4, (c) Total T3 and (d) Free T3 levels A significant reduction in the values of free and total thyroid hormones T4 and T3 was brought about by the administration of CYP (400 mg/kg, intraperitoneal injection). However, the administration of MET did not have any effect on the levels of both free as well as total T4 or T3. In contrast to the saline group, the CYP+MET group did not demonstrate any indication of improvement in either the T4 or T3 levels. The SEM is used to express the data. ***p<0.001 and ****p<0.0001 when it compared with animals that were given saline

concentrations of T4 and T3 in both their free and total forms were significantly lower in the animal groupings that was treated with CYP in comparison to those given saline. The T4 and T3 concentrations were not improved by the co-treatment with MET; rather, they were decreased by the treatment with CYP (Fig. 3a-d).

DISCUSSION

The principal purpose of current research was to investigate the influence of MET on animals that had been given treatment for CYP-induced hypothyroidism. The TSH, Free Thyroxine (FT4), total Triiodothyronine (T3), Free Triiodothyronine (FT3) and total Thyroxine (T4) were the thyroid hormones that were the focus of this study. The rationale of this research was to estimate the influence that CYP has on these hormones. Additionally, the objective also included to determine whether or not MET could have a beneficial effect on thyroid function by mitigating the negative effects of CYP-induced toxicity.

The Kaplan-Meier method is utilized quite frequently in the process of analyzing data pertaining to the time between events²². This method proves to be valuable in survival/endurance rate analysis, as it enables investigators to assess and examine the attrition of test subjects in a particular study²³. It is commonly employed to evaluate and contrast two sets of participants, such as a placebo grouping and a control grouping or another treatment/induced grouping²³. Thus, the effects of CYP and MET, as well as their combination, were evaluated in rat models in the present study. The findings indicated a decline in the survival rate observed in the group that received CYP treatment. Furthermore, a greater number of rats exhibited a decrease in survival when MET was administered in conjunction with CYP.

There is a correlation between the existence of hypothyroidism and alterations in a number of physiological activities²⁴. These alterations can be linked to the distribution of thyroid receptor (THR) function²⁵. Several varieties of THRs, known as THR α and TR β are identified, which can be found in various organs such as the brain and the heart. The THRs are crucial in the regulation of gene transcription, operating within the nucleus⁵. Typically, the concentrations of TSH and thyroid hormones (T4 and T3) in the bloodstream can provide insight into changes in thyroid hormone concentrations and function, like in hypothyroidism and hyperthyroidism²⁶. For instance, the decrease in T4 and T3 concentrations, along with an increase in TSH, indicates the presence of primary hypothyroidism^{26,27}. The decline in the concentrations of thyroid hormonal levels in the bloodstream can have an

impact on the activity of THR, potentially leading to changes in the functioning of the brain and heart^{27,28}.

The concentrations of T3, T4, TSH, FT4 and FT3 were examined in the study following treatment with CYP, MET, as well as mixture of CYP with MET. The outcomes of the study showed non-significant changes in TSH concentrations across all treated groups. Significant reductions in T4, FT4, T3 and FT3 concentrations were observed in rats treated with CYP and CYP+MET, as linked to the saline/untreated group. Nevertheless, the levels of T4 and FT4 remained unchanged in rats treated with MET when compared to those treated with saline. According to these findings, it has been discovered that CYP has the ability to impact thyroid function by influencing the concentrations of thyroid hormones. This, in turn, can have an effect on the activity of TR and the proper functioning of organs that rely on thyroid hormones.

CONCLUSION

The treatment with CYP can lead to the development of hypothyroidism, resulting in alterations to various regulatory physiological functions. The study proposed that CYP may cause hypothyroidism by affecting the concentrations of circulating TSH, total and free T4 and T3. Therefore, co-treatment with MET could potentially alleviate the impact of CYP on hypothyroidism. The findings revealed that in rats with CYP-induced hypothyroidism, MET did not alleviate the toxic impact of CYP and did not enhance the significant reduction in the levels of total and free T4 as well as T3.

SIGNIFICANCE STATEMENT

This research aims to estimate the potential amelioratory effects of metformin against hypothyroidism induced by cyclophosphamide. In a previous study, it was discovered that cyclophosphamide can cause a notable reduction in the concentrations of total and free Triiodothyronine (T3) and Thyroxine (T4), leading to hypothyroidism. According to the research, rats that received metformin did not show any enhancements in their T3 and T4 concentrations. The results suggest that metformin does not improve hypothyroidism caused by cyclophosphamide.

REFERENCES

 Fröhlich, E. and R. Wahl, 2021. Physiological role and use of thyroid hormone metabolites-potential utility in COVID-19 patients. Front. Endocrinol., Vol. 12. 10.3389/fendo.2021.587518.

- 2. Silva, J.F., N.M. Ocarino and R. Serakides, 2018. Thyroid hormones and female reproduction. Biol. Reprod., 99: 907-921.
- 3. Gregory A. Brent., 2012. Mechanisms of thyroid hormone action. J. Clin. Invest., 122: 3035-3043.
- 4. McAninch, E.A. and A.C. Bianco, 2016. The history and future of treatment of hypothyroidism. Ann. Intern. Med., 164:50-56.
- 5. Ortiga-Carvalho, T.M., A.R. Sidhaye and F.E. Wondisford, 2014. Thyroid hormone receptors and resistance to thyroid hormone disorders. Nat. Rev. Endocrinol., 10: 582-591.
- 6. Mayerl, S., H. Heuer and C. Ffrench-Constant, 2020. Hippocampal neurogenesis requires cell-autonomous thyroid hormone signaling. Stem Cell Rep., 14: 845-860.
- 7. Baksi, S. and A. Pradhan, 2021. Thyroid hormone: Sexdependent role in nervous system regulation and disease. Biol. Sex. Differ., Vol. 12. 10.1186/s13293-021-00367-2.
- 8. Samuels, M.H., 2014. Psychiatric and cognitive manifestations of hypothyroidism. Curr. Opin. Endocrinol. Diabetes Obesity, 21: 377-383.
- Dhesi, S., M.P. Chu, G. Blevins, I. Paterson, L. Larratt, G.Y. Oudit and D.H. Kim, 2013. Cyclophosphamide-induced cardiomyopathy: A case report, review, and recommendations for management. J. Invest. Med. High Impact Case Rep., Vol. 1. 10.1177/2324709613480346.
- 10. de Montellano, P.R.O., 2013. Cytochrome P450-activated prodrugs. Future Med. Chem., 5: 213-228.
- 11. Baiken, Y., D. Kanayeva, S. Taipakova, R. Groisman and A.A. Ishchenko *et al.*, 2020. Role of base excision repair pathway in the processing of complex DNA damage generated by oxidative stress and anticancer drugs. Front. Cell Dev. Biol., Vol. 8. 10.3389/fcell.2020.617884.
- 12. Shi, H., B. Hou, H. Li, H. Zhou and B. Du, 2022. Cyclophosphamide induces the ferroptosis of tumor cells through heme oxygenase-1. Front. Pharmacol., Vol. 13. 10.3389/fphar.2022.839464.
- Zhang, L., Y. Shi, J. Zhang, J. Wu and W. Jiang, 2023. Cyclophosphamide-induced seizures in a patient with neuropsychiatric systemic lupus erythematosus (NPSLE): A case report. Front. Immunol., Vol. 14. 10.3389/fimmu.2023.1122629.
- Teles, K.A., P. Medeiros-Souza, F.A.C. Lima, B.G. de Araújo and R.A.C. Lima, 2017. Cyclophosphamide administration routine in autoimmune rheumatic diseases: A review. Rev. Bras. Reumatologia, 57: 596-604.
- 15. Lee, K.O., I. Kwon, H.S. Nam, Y. Park and J. Kim *et al.*, 2021. Effect of leukopenia induced by cyclophosphamide on the initial stage of arterial thrombosis in mice. Thromb. Res., 206: 111-119.

- 16. Ferreira, M.N., J.Y. Ramseier and J.S. Leventhal, 2019. Dermatologic conditions in women receiving systemic cancer therapy. Int. J. Women's Dermatol., 5: 285-307.
- Deutsch, A., N.R. Leboeuf, M.E. Lacouture and B.N. McLellan, 2020. Dermatologic adverse events of systemic anticancer therapies: Cytotoxic chemotherapy, targeted therapy, and immunotherapy. Am. Soc. Clin. Oncol. Educ. Book, 40: 485-500.
- Lv, Z. and Y. Guo, 2020. Metformin and its benefits for various diseases. Front. Endocrinol. Vol. 11. 10.3389/fendo.2020.00191.
- Alhowail, A., 2021. Potential mechanisms of metformininduced memory impairment. Eur. Rev. Med. Pharmacol. Sci., 25: 4757-4761.
- 20. Zhou, W., A. Kavelaars and C.J. Heijnen, 2016. Metformin prevents cisplatin-induced cognitive impairment and brain damage in mice. PloS One, Vol. 11. 10.1371/journal.pone.0151890.
- Alotayk, L.I., M.A. Aldubayan, S.K. Alenezi, M.J. Anwar and A.H. Alhowail, 2023. Comparative evaluation of doxorubicin, cyclophosphamide, 5-fluorouracil, and cisplatin on cognitive dysfunction in rats: Delineating the role of inflammation of hippocampal neurons and hypothyroidism. Biomed. Pharmacother., Vol. 165. 10.1016/j.biopha.2023.115245.
- D'Arrigo, G., D. Leonardis, S. Abd ElHafeez, M. Fusaro, G. Tripepi and S. Roumeliotis, 2021. Methods to analyse timeto-event data: The Kaplan-Meier survival curve. Oxid. Med. Cell. Longevity, Vol. 2021. 10.1155/2021/2290120.
- Dudley, W.N., R. Wickham and N. Coombs, 2016. An introduction to survival statistics: Kaplan-Meier analysis. J. Adv. Pract. Oncol., 7: 91-100.
- 24. Chaker, L., S. Razvi, I.M. Bensenor, F. Azizi, E.N. Pearce and R.P. Peeters, 2022. Hypothyroidism. Nat. Rev. Dis. Primers, Vol. 8. 10.1038/s41572-022-00357-7.
- Liu, Y.C., C.T. Yeh and K.H. Lin, 2019. Molecular functions of thyroid hormone signaling in regulation of cancer progression and anti-apoptosis. Int. J. Mol. Sci., Vol. 20. 10.3390/ijms20204986.
- Razvi, S., S. Bhana and S. Mrabeti, 2019. Challenges in interpreting thyroid stimulating hormone results in the diagnosis of thyroid dysfunction. J. Thyroid Res., Vol. 2019. 10.1155/2019/4106816.
- 27. Udovcic, M., R.H. Pena, B. Patham, L. Tabatabai and A. Kansara, 2017. Hypothyroidism and the heart. Methodist DeBakey Cardiovasc. J., 13: 55-59.
- 28. Cooke, G.E., S. Mullally, N. Correia, S.M. O'Mara and J. Gibney, 2014. Hippocampal volume is decreased in adults with hypothyroidism. Thyroid, 24: 433-440.